Author's response to reviews

Title: Child survival: A message of hope but a call for renewed commitment in UNICEF report

Authors:

Tessa Wardlaw (twardlaw@unicef.org)
Danzhen You (dyou@unicef.org)
Holly Newby (hnewby@unicef.org)
David Anthony (danthony@unicef.org)
Mickey Chopra (mchopra@unicef.org)

Version: 2
Date: 30 November 2013

Author's response to reviews:

Dear Regie,

Thank you for your email. I have changed the following for our manuscript as you advised and resubmitted online:

• Included a title page as page 1.
• Included the Abstract on page 2. – As this is just a commentary, we did not specify background, methods, results and conclusions. If you still need that, please let us know.
• Included a competing interests section at the end of the manuscript.
• Included an acknowledgement section at the end of the manuscript before the reference list – note that we don’t have anyone to acknowledge for this short commentary.
• Included a list of abbreviations – not that we don’t have abbreviations in this commentary.
• Include figure title and legend section after the reference list.

I also attached the revised manuscript here. Please let us know whether additional changes are requested.

Thank you very much.

Best regards,

Danzhen